**Author’s response to reviews**

**Title:** Association of ABO Polymorphisms and Pancreatic Cancer/ Cardiocerebrovascular Disease: a Meta-analysis

**Authors:**

Lian Li (lilian@tmu.edu.cn)

Yanxia Li (liyanxia0525@163.com)

Luyang Liu (iuluyang0628@163.com)

Yubei Huang (yubei_huang@163.com)

Hong Zheng (zhengh64@aliyun.com)

**Version:** 4  **Date:** 12 Feb 2020

**Author’s response to reviews:**

1. **Table 3**

   We note that you have included an in-text citation to “Table 3”, however this table is absent from your manuscript. If you wish to include this citation, please upload the table into your manuscript and into the file inventory. However, if you do not wish to include this table in your manuscript, please remove the in-text citation corresponding to “Table 3” on page 10 of your manuscript.

   **Response:** We thank the editor for catching this error. We have remove the “Table 3” on page 10 and edited the sentence with a new sentence in lines 11-13, page 10 in the revised manuscript, as follows: “We conducted Begg’s and Egger’s tests to assess the publication bias for these studies and no evidence of publication bias was detected among the enrolled studies (Fig. 4c; Table S4).”. Thank you for helping us to improve the quality of our manuscript.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

   **Response:** We have uploaded a clean version of main text, tables, figures, and additional files in this version of manuscript.